Author’s response to reviews

Title: Early Results of a Novel Technique for Anterior Vaginal Wall Prolapse Repair: Anterior Vaginal Wall Darn

Authors:

Osman Köse (koseonk@yahoo.com.tr)
Hasan S Saglam (hasanss@hotmail.com)
Sükrü Kumsar (drkumsar@yahoo.com)
Salih Budak (salihbudak1977@gmail.com)
Hüseyin Aydemir (husaydemir@yahoo.com)
Öztug Adsan (oztugadsan@yahoo.com)

Version: 4 Date: 11 June 2014

Author’s response to reviews: see over
June 11, 2014

Dear Editor;

Thank you very much for your assistance in making out our manuscript better. We have written our corrections below according to your suggestions.

1. The full name of the ethics committee has been added to page 4, lines: 3-4.
2. The patient’s consent to publish her image in figure 3 has been written in the figure legend at page 11, lines:19-20.
3. The copyright of the figures belongs to our clinic database. We have given permission for these to be reproduced in a BioMed Central publication.
4. We have uploaded our abstract again.

Best regards,

Osman Köse